Author's response to reviews

Title: Computer-assisted versus non-computer-assisted preoperative planning of corrective osteotomy for extra-articular distal radius malunions: a randomized controlled trial

Authors:

Natalie L Leong (natalie_leong@hms.harvard.edu)
Geert A Buijze (gbuijze@partners.org)
Eric C Fu (efu@partners.org)
Filip Stockmans (Filip.Stockmans@kuleuven-kortrijk.be)
Jesse B Jupiter (jjupiter1@partners.org)

the Distal Radius Malunion (DiRaM) collaborative group
(jjupiter1@partners.org)

Version: 3 Date: 8 December 2010

Author's response to reviews: see over
December 8, 2010

Dear Editor:

Attached please find a copy of the manuscript (ID 2069809068439635) entitled “Computer-assisted versus non-computer-assisted preoperative planning of corrective osteotomy for extra-articular distal radius malunions: a randomized controlled trial” by Leong N. et al., with editorial changes in preparation for publication in BMC Musculoskeletal Disorders. In accordance with request by the editorial production team, the following changes have been made to the manuscript:

Major revisions:

1) The author names and author order has been changed such that they match exactly in the manuscript and the submission system.

2) The names of members of the study group have been placed in a new “Investigators and Participating Centers” section.

3) The abstract has been modified such that the versions in the manuscript and the submission match exactly.

Minor revisions:

1) The title has been changed to sentence case in both the manuscript and the submission system.

2) The author names, affiliations, and email addresses have been separated on the title page.

3) The heading “Abstract” has been added to the abstract.

4) The section titled 'Introduction' in the abstract has been renamed 'Background'.

5) The methods section has been renamed ‘Methods/Design’.

6) The formatting of the reference list has been standardized, and the full list of author names has been used for each publication rather than the term ‘et. al.”

7) The entire manuscript has been proofread for typographical errors.

Please do not hesitate to contact me if any additional information is required.

Best regards,

Natalie Leong
Massachusetts General Hospital
natalie_leong@hms.harvard.edu